Author’s response to reviews

Title: Impact of an integrated care program on glycemic control and cardiovascular risk factors in patients with type 2 diabetes in Saudi Arabia: An interventional parallel-group controlled study

Authors:

Ayla Tourkmani (aylatourkmani@gmail.com)
Osama Abdelhay (osamaabdelhay@gmail.com)
Hesham Alkhashan (drh868@yahoo.com)
Aboud Alaboud (aalabood@psmmc.med.sa)
Ahmed Bakhit (drahbakhiet@hotmail.com)
Tarek Elsaid (tarek_elsaid@hotmail.com)
Ahmed Alawad (aalawad@rmh.med.sa)
Aljohara Alobaikan (jalobaikan@psmmc.med.sa)
Hala Alqahtani (halahqahtani@yahoo.com)
Abdulaziz Alqahtani (abuhanin82@hotmail.com)
Adel Mishriky (adel.mishriky@gmail.com)
Abdulaziz Bin Rsheed (ambirrasheed@psmmc.med.sa)
Turki Alharbi (Turkifcm@yahoo.com)

Version: 3 Date: 07 Dec 2017

Author’s response to reviews:

Point-by-point responses

Thank you for your valuable suggestions. According to reviewers' comments, we carefully revised our final version of the manuscript.

To the Editor

Thank you for your review and comments.
1. Including trial registration at the end of the manuscript's abstract.
The trial registration ID has been included at the end of the abstract per the journals guideline.

2. Authors contribution:
   
   a. The full authors names are included rather than their initials (page 16).
   
   b) Each co-author has been assigned with their contribution that qualify them to be in co-authorship as per the journal policy (page 16)
   
   c) Dr. Abdulaziz bin Rasheed is now included in the authors contribution section.

3. Availability of the raw data

The availability of the raw data statement has been changed to adhere with the journal’s guidelines (page 16)

4. Citation of Table 1 and Supplementary Table 1.

- Table 1 was cited in the text page 9 in the previous versions and it is still in the current version of the manuscript.

- Supplementary Table 1 is now cited in page 8.

5. Figures

Each figure is now attached in a separate file as instructed with additional files section at the end of the manuscript has been added (page 29).

Kind Regards

Dr. Turki Alharbi